Author’s response to reviews

Title: Effects and outcomes of septostomy in twin-to-twin transfusion syndrome after fetoscopic laser therapy

Authors:

Wen-Fang Li (wanna0507@gmail.com)
An-Shine Chao (aschao1295@cgmh.org.tw)
Shuenn-Dyh Chang (gene@cgmh.org.tw)
Po-Jen Cheng (pjcheng@cgmh.org.tw)
Lan-Yan Yang (lyyang0111@gmail.com)
YL Chang (j12054@cgmh.org.tw)

Version: 2 Date: 02 Oct 2019

Author’s response to reviews:

Reviewer reports:

Stephanie Springer (Reviewer 1):

Please revise Page 9 Line 14: For about 23.7 million populations, there are currently two laser therapy centers for TTTS in Taiwan

Answer: thanks, we had changed the sentences as” Only two laser therapy centers for TTTS currently serve the 23.7 million residents of Taiwan. Ours is the larger laser therapy center and has handled more than 95% of cases.”

Results: The Editorial Board should decide if you really should write e.g. 7 cases in parenthesis beneath the percentages. I preferred the older version.

Answer: We had changed it to the older version.

The Editorial Board should decide if Table 1 has the right format.

authors.
David Alan Ellwood (Reviewer 2): Thanks for responding to my comments and questions. I am grateful for the revisions made.

Answer: thanks.

Maria Delius, MD, MPH (Reviewer 3): Dear authors, thank you very much for the quick corrections. Please let the newly incorporated sentences be corrected by a native Speaker.

Thank you.

Answer: we had corrected those sentences accordingly.